Author’s response to reviews

Title: Effect of Ketofol versus propofol as an induction agent on ease of laryngeal mask airway insertion conditions and hemodynamic stability in pediatrics: An observational prospective cohort study.

Authors:

Bacha Aberra (bcabera11@gmail.com)
Adugna Aregawi (adugenet@gmail.com)
Girmay Teklay (girmeat@gmail.com)
Hagos Tasew (tasewh2@gmail.com)

Version: 1 Date: 07 May 2018

Author’s response to reviews:

Thank you for the your constructed comment

We submit a clean version of our manuscript and the track changes or highlighting are modified.

we made revision on the main manuscript and most of the overlapped texts are from my original thesis in Addis Abbaba university repository library.

we already resubmit our clean version manuscript without the actual trial registration number but registration is in progress start to register retrospectively. We can add the trial registration number in the manuscript at a later time.